Author's response to reviews

Title: Etiological study of enteric viruses and the genetic diversity of norovirus, sapovirus, adenovirus, and astrovirus in children with diarrhea in Chongqing, China

Authors:

Zengzhi Ren (renzengzhi@gmail.com)
Yuanmei Kong (kongyuanmei@163.com)
Jun Wang (followbeyond@yahoo.com.cn)
Qianqian Wang (cherish_119@126.com)
Ailong Huang (ahuang1964@yahoo.com.cn)
Hongmei Xu (xuhongmwpx@yahoo.com)

Version: 3 Date: 26 January 2013

Author's response to reviews:

Dear editor-in-Chief#

According to Editorial Requirements, we have added the information requested.

1. Requesting ethics statement: It was added in the first paragraph in the Methods section.

2. Requesting consent statement: It was also added in the first paragraph in the Methods section.

Best Wishes!

Hongmei